Author’s response to reviews

Title: Retrospective analysis of seven cases of pancreatic mixed adenoneuroendocrine carcinoma from a high-volume center and review of the literature

Authors:
Feng Tian (andytianfeng@126.com)
Menghua Dai (daimh@pumch.cn)
Cong-Wei Jia (david_jia0814@163.com)
Zi-Wen Liu (lzw1999@sohu.com)
Bing-Lu Li (libinglu@pumch.cn)

Version: 4 Date: 22 Jun 2019

Author’s response to reviews:
Dear Editors,

Thank you very much for your final positive and exciting reply. We appreciated the editor’s and the reviewers’ helpful comments and important suggestions, which would benefit us in better conducting a scientific study or preparing a manuscript in the future. We’ve learned a lot.

We have finished minor revisions as requested by the editor: expression change in ethics approval and consent to participate section (line 227-228, page 8), and represent the authors’ names using initials in the Authors’ Contributions section (line 244-246, page 8).

Thanks again and look forward to the final acceptance by your journal.

Yours sincerely,
Dr. Meng-hua Dai

Point-to-point answers:
Riccardo Casadei (Reviewer 1): THE MANUSCRIPT REVISED IS VERY GOOD AND IT CAN BE ACCEPTED
Response: Thank you so much for the exciting comment! Best wishes!